Author’s response to reviews

Title: Infant feeding practice and associated factors of HIV positive mothers attending prevention of mother to child transmission and antiretroviral therapy clinics in Gondar Town health institutions, Northwest Ethiopia

Authors:

  Dagnachew F Muluye (fetenemulu24@gmail.com)
  Desalegn K Woldeyohannes (desalegn01@gmail.com)
  Mucheye B Gizachew (muchegiza@gmail.com)
  Moges W Tiruneh (mogest4@gmail.com)

Version: 6 Date: 4 March 2012

Author’s response to reviews:

Dear Sir,

For the comment on Study design, area and period section “estimated HIV prevalence of 2.8% in adult women (2009)”;
- It was the single point estimated HIV prevalence of Ethiopia in 2009 but we agreed that is not relevant and we have deleted it.